Author's response to reviews

Title: Inter-examiner reproducibility of tests for lumbar motor control

Authors:

Flemming Enoch (fe@fysiq.dk)
Per Kjaer (pkjaer@health.sdu.dk)
Arne Elkjaer (arne.e@mail.dk)
Lars Remvig (lars.remvig@rh.regionh.dk)
Birgit Juul-Kristensen (bjuul-kristensen@health.sdu.dk)

Version: 5 Date: 11 May 2011

Author's response to reviews:

The canges made ar:
-typing errors corrected and equalizing script formats
-elimination of track changes
-new tables 2, 3, 4 as requested and a revised figure 2
-moving of tables and figures
-removal of the envelope at title page